Title: Safety, efficiency and health-related quality of telephone triage conducted by general practitioners, nurses, or physicians in out-of-hours primary care: a quasi-experimental study using the Assessment of Quality in Telephone Triage (AQTT) to assess audio-recorded telephone calls

Authors:

Dennis Graversen (d.graversen@ph.au.dk)
Morten Christensen (mbc@ph.au.dk)
Anette Pedersen (afp@ph.au.dk)
Anders Carlsen (andershelles@gmail.com)
Flemming Bro (fbro@ph.au.dk)
Helle Christensen (helle.collatz.christensen.02@regionh.dk)
Claus Høstrup Vestergaard (cv@ph.au.dk)
Linda Huibers (huibers@ph.au.dk)

Version: 4 Date: 02 Mar 2020

Author’s response to reviews:

Dear Marie Clayton and Gerard Clunn,

We are pleased to hear that you find our paper potentially acceptable for publication.

We have;

1: Moved the keywords after the abstract

2: Removed the marked version

3: Uploaded a clean version, in which we have removed the two supplementary files (Appendix 1 &amp; 2) and uploaded them separately.

We hope you will find that we have made the required adjustments.

Best regards on behalf of the authors,
Dennis Schou Graversen